Author's response to reviews

Title: Family history and stroke outcome in a bi-ethnic, population-based stroke surveillance study

Authors:

Lynda D Lisabeth (llisabet@umich.edu)
Melinda A Smith (melindac@med.umich.edu)
Devin L Brown (devinb@med.umich.edu)
Ken Uchino (uchinok@upmc.edu)
Lewis B Morgenstern (LMorgens@umich.edu)

Version: 4 Date: 11 October 2005

Author's response to reviews:

October 12, 2005

Dear Editor,

We are submitting the final version of our manuscript entitled "Family history and stroke outcome in a bi-ethnic, population-based stroke surveillance study" electronically. The following formatting changes were made to the manuscript.

1. Authors' names have been checked.
2. Abstract/title have been checked.
3. All references have been placed before the full stop.
4. Highlighting has been removed.
5. References contain all author names regardless of number.
6. Table titles have been changed to sentence case.
7. Other small editorial changes were also made upon final review.

Sincerely,

Lynda Lisabeth, PhD